Author's response to reviews

Title: Possible Transmission of HIV-1 Infection from an Elite Controller to a Patient who Progressed to AIDS: a case report

Authors:

Michael S Killian (scott.killian@ucsf.edu)
Girish N Vyas (girish.vyas@ucsf.edu)
Rochak Mehta (rochak.mehta@roche.com)
Karen Young (karen.young@roche.com)
Osman Ebrahim (osman.ebrahim@telkomsa.net)

Version: 6 Date: 22 May 2012

Author's response to reviews:

To address the Deputy Editor’s concerns, we have now added five instances of the word ‘possible’ and three uses of the word ‘possibly’. The title and the conclusion have been modified as suggested and each claim about transmission is now associated with a form of the word ‘possible’. We hope that the revised manuscript is satisfactory and can now be accepted for publication in Journal of Medical Case Reports.

Thank you.

Scott Killian and co-authors